Author's response to reviews

Title: The genetic susceptibility to type 2 diabetes may be modulated by obesity status: implications for association studies

Authors:

Stéphane Cauchi (Stephane.Cauchi@good.ibl.fr)
Kevin T Nead (knuf08@ufl.edu)
Hélène Choquet (helene@good.ibl.fr)
Fritz Horber (fritz@horber.org)
Natascha Potoczna (natascha.potoczna@horber.org)
Beverley Balkau (balkau@vjf.inserm.fr)
Michel Marre (michel.marre@bch.ap-hop-paris.fr)
Guillaume Charpentier (guillaume.charpentier@ch-sud-francilien.fr)
Philippe Froguel (P.Froguel@imperial.ac.uk)
David Meyre (meyre@good.ibl.fr)

Version: 4 Date: 14 May 2008

Author's response to reviews: see over
Response to Editor (Manuscript ID 1095391601180803)

Q1: Please present all text in black.

R1: We unmarked our previous corrections and all the text is now presented in black.

Q2: Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

R2: We made the appropriate changes

Q3: Qualifications - Please remove the authors' qualifications (e.g. PhD / MD) from the manuscript and submissions system

R3: We made the appropriate changes

Q4: Email and affiliations - Please separate the email addresses and the affiliations. Each email address should be on a new line and should be preceded by the author's initials, not the full name. Affiliations - Please write each affiliation on a new line.

R4: We made the appropriate changes
Q5: Please present ‘Abbreviations’ as a paragraph rather than a list.

R5: In the instructions to authors, it is stated that abbreviations can be defined in a list. However, we changed the paragraph marks for semicolons. Please let us know if it was what you requested.

Q6: Endnote - There is a discrepancy between the number of references in your manuscript reference list, and the number extracted by our system. This can be seen on the upload page of the submission system in the main manuscript box. For help please visit: http://www.biomedcentral.com/info/ifora/endnote/#technicalhelp

R6: We noticed that one reference was indeed missing:


We followed your suggestions and did not find where the problem came from. We contacted: references@biomedcentral.com. We did not have any news from them. Please let us know what was wrong with our manuscript.

Q7: Tables size ¿ These must fit into an A4 portrait page.

R7: We made the appropriate changes

Q8: Additional files ¿ All additional files should be mentioned in the text, and referred to in the following way: ‘[see Additional file 1]’. 
R8: We made the appropriate changes. (Supplementary Table n) was changed for [see Additional file n]

Please let us know if you need anything else.

Best Regards,

Stéphane Cauchi and Philippe Froguel